Author's response to reviews

Title: Cardiac arrest provoked by itraconazole and amiodarone interaction: a case report

Authors:

Angeliki M Tsimogianni Dr (atsimog@med.uoa.gr)
Alex Betrosian Dr (abetrosian@gmail.com)
Emmanouil Douzinas Dr (edouzin@cc.uoa.gr)
Ilias Andrianakis Dr (iandrianakis@yahoo.gr)

Version: 2 Date: 19 October 2010

Author's response to reviews: see over
Dear Editor of Journal of Medical Case Reports,

We are hereby submitting our revised manuscript entitled: “Cardiac Arrest Provoked By Itraconazole And Amiodarone Interaction: a case report” for consideration for publication in the Journal as a Case Report.

We have tried to adjust to the instructions for authors and we have made all changes suggested by the Editorial Support team. In specifically we have amended the title page, removed the information about the institution where the work was performed, we separated the conclusion from the discussion session and we amended the consent form as suggested. We want also declare that the material is original and is not currently under consideration for publication nor has been accepted for publication elsewhere. All authors have contributed significantly to the work, read and approved the submitted version of the paper. There are not related papers submitted for publication. All authors declare that they have no conflict of interest. All data represent honest work.

In this study, we describe a case of itraconazole induced cardiac arrest in a patient with ischemic stroke and atrial fibrillation controlled on amiodarone. This is the first case reported of cardiac arrest caused by interaction of itraconazole and amiodarone. Intensivists and Medical Physicians of many specialties like Cardiologists, Oncologists and Haematologists should be aware of this interaction. This patient had also an exceptional clinical course: he survived from an extensive cerebrovascular accident,
recovered from septic shock complicated with multiorgan failure, had another extensive stroke and was resuscitated successfully from 2 cardiac arrests.

Thank you very much in advance for critically appraising our manuscript.

Sincerely,

Angeliki Tsimogianni, MD

Registrar in Respiratory and Critical Care Medicine,

Home address: 113 Peloponnisou Street, 16451, Argyroupolis, Athens, Greece; Tel: +30 210 9944013, +30 6944546132,

Work address: Evgenidion Hospital, School of Medicine, National and Kapodistrian University of Athens, 20 Papadiamantopoulou street, 11528 Athens, Greece.

Tel: +30 210 7208116; Fax: +30 210 7220540; e-mail: atsimog@med.uoa.gr.